Author's response to reviews

Title: Family physician and endocrinologist coordination as the basis for diabetes care in clinical practice

Authors:

Alejandra Duran (aduranh@hotmail.com)
Isabelle Runkle (irunkle.hcsc@salud.madrid.org)
Pilar Matia (mmatia.hcsc@salud.madrid.org)
Paz de Miguel (mdemiguel.hcsc@salud.madrid.org)
Sofia Garrido (sgarrido.gapm07@salud.madrid.org)
Emilio Cervera (ecervera.gapm07@salud.madrid.org)
Dolores Fernandez (mdfernandez.hcsc@salud.madrid.org)
Pilar Torres (mtorres.hcsc@salud.madrid.org)
Tomas Lillo (tilllo.hcsc@salud.madrid.org)
Patricia Martin (pmartinrojasmarcos@yahoo.es)
Lucio Cabrerizo (lcberrerizo.hcsc@salud.madrid.org)
Nuria Garcia de la Torre (nurialobo@hotmail.com)
Jose R Calle (ijrcalle@telefonica.net)
Jose Ibarra (jjibarra@hotmail.com)
Aniceto L Charro (acharro.hcsc@salud.madrid.org)
Alfonso L Calle-Pascual (acalle.hcsc@salud.madrid.org)

Version: 11 Date: 28 July 2008

Author's response to reviews: see over
Dear Sir,
I send you the revised form of the pre-accept m.n. FAMILY PHYSICIAN AND ENDOCRINOLOGIST COORDINATION……., to BMC Endocrine Disorder, according to formatting requirements. I had holidays the lasts two weeks.
The following formatting changes have been performed:
- Keywords, word-count and running title have been deleted.
- Author list in the m.s. has been written exactly as in the submissions system.
  Qualifications and titles have been removed.
- Affiliations have been written in full.
- Abbreviations are included after the conclusions section
- Unnecessary capitals have been removed from section headings
- Abstract has been structured as indicated
- Introduction has been renamed background.
- Authors contributions has been changed to Authors´ contributions
- All blank pages have been removed
- Figure has been deleted in the m.n. file, and will be upload as separate on the submission system.
- Website address has been included in the availability section.

Sorry again for send you the revised form today after my holidays.
With best wishes,

Alfonso L. Calle-Pascual